CSE 326:

Data Structu res

Spring 1996

Course Information

Instructor: Martin T ompa Sieg 426B Mon 3:00 { 4:00

tompa@cs.w ashin gton.edu 543-9263 W ed 2:00 { 3:00

T A: Juan Aleman y Sieg 326D Th u 10:30 { 11:30

juanito@cs.w ashington.ed u

Prerequisites: CSE 321.

Grading: Homew ork (40%), Midterm (20%), Final (40%). P ercen tages are appro ximate.

Computing facilities: Sieg 329 Unix lab, C++

Course W eb: http:// www. cs.w ashi ngt on.e du/e duca tio n/co urse s/cs e326 /

T extb o ok: Harry Lewis and Larry Denen b erg, Data Structur es & Their A lgorithms,

Harp erCollins, 1991.

Reading assignme n t: P ages 1{25, b y next lecture. The sections on \Notation for Pro-

grams", \Lo cativ es", \Abstract Data T yp es ", and \Order Notation" are particularly cen tral

to this course.